Author's response to reviews

Title: Emergency Department Triage: an Ethical Analysis

Authors:

  Ramesh P Aacharya (raacharya@healthnet.org.np)
  Chris Gastmans (Chris.Gastmans@med.kuleuven.be)
  Yvonne Denier (Yvonne.Denier@med.kuleuven.be)

Version: 4 Date: 21 August 2011

Author's response to reviews: see over
Date: 18 August 2011

The Chief Editor
BMC Emergency Medicine

Subject: Submission of revised manuscript.

Dear Sir,
I hereby, submit the revised manuscript addressing the reviewer’s comments to the article entitled “Emergency Department Triage: an Ethical Analysis” co-authored by me, Prof. Chris Gastmans and Dr. Yvonne Denier.

1. The abstract is restructured with separate subheadings of background, discussion and summary.
2. As suggested by referee 2, START triage algorithm is added as Figure 3.

Along with this cover letter, I am enclosing the revised manuscript with track changes.

If any queries, kindly let me know. Thank you very much for your kind attention and cooperation.

Sincerely,

Ramesh Prasad Aacharya
Department of General Practice & Emergency Medicine,
Tribhuvan University Teaching Hospital, Maharajgunj,
P. O. Box 8844, Kathmandu, NEPAL
Email: raacharya@healthnet.org.np